RESEARCH Open Access

Improving adult inpatient eating disorder treatment: perspectives of a sample of individuals in Canada with lived experience



Catherine Quinn Armour^{1*}, Becky Feicht¹ and Jacqueline Gahagan²

Abstract

Background Eating disorders (EDs) are increasingly common, affecting over one million individuals in Canada. Canadian adults (i.e., age 18+) requiring medical stabilization for their eating disorder (ED) may be referred to adult inpatient (IP) ED treatment for care. Recent Canadian publications have brought attention to the need for improved approaches to Canadian ED treatment; urging researchers to seek perspectives of people with lived experience to determine how to best do so. This study explored the perspectives of a sample of individuals in Canada with lived experience to identify recommendations for improvement of adult IP ED treatment programs and processes.

Methods Employing a transformative philosophical view and feminist standpoint theory, this study utilizes a qualitative hermeneutic phenomenological approach to fulfill the objectives. Eleven participants with lived adult IP ED treatment experiences from across Canada were interviewed individually, to discuss their experiences and recommendations regarding referral, transitions into and out of care, and treatment itself using an online video conference platform. Data were analyzed using interpretative phenomenological analysis. A comprehensive list of recommendations was drafted and brought back to participants for feedback. The feedback was implemented to create the final list of recommendations.

Results Several limitations of referral, transitions, and treatment, facilitated and exacerbated by stigma at individual and societal levels, were identified by participants. These included guilt and shame upon referral, lack of respect and trust from healthcare providers during transitions, and lack of consideration of social determinants of health during treatment. Participant-informed recommendations, which can be categorized as *interim support*, *individualized care*, *dignified treatment*, *resources*, and *stigma*, were identified to ameliorate the experiences of those in Canada with EDs while also combatting stigma.

Conclusions Adult IP ED treatment in Canada is in urgent need of significant change to meet the needs of those requiring care and to address harmful stigma. Implementing participant-informed recommendations may aid in achieving this goal. The meaningful inclusion of those with lived experience, particularly marginalized populations, will be paramount to the development of an approach to adult IP ED treatment that properly serves individuals in Canada who need it.

*Correspondence: Catherine Quinn Armour MA catherine.armour@dal.ca

Full list of author information is available at the end of the article



Plain English Summary

Eating disorders (EDs) are common and can be deadly, yet treatment approaches are not currently meeting the needs of individuals in Canada requiring care. As noted in the existing literature, ED treatment is largely inaccessible in Canada and can be ineffective in facilitating long-term recovery. Recent research has suggested that improvement to publicly funded, hospital-based ED treatment for adults is necessary and that those in Canada with lived experience should be involved in the process of doing so. This paper is based on a study that set out to explore ED treatment and gather potential recommendations from eleven individuals in Canada with lived experience. Several recommendations were identified to improve ED treatment in Canada, while simultaneously addressing harmful stigma about people with lived experience with EDs. These recommendations included providing interim support, ensuring individualized care, offering dignified treatment, increasing resources, and directly targeting stigma. From this study, it is suggested that current ED treatment methods must change to improve health outcomes and reduce stigma, and exploring the perspectives of those with lived experience is a promising method for doing so.

Keywords Eating disorders, Inpatient treatment, Adults, Lived experience, Recommendations, Stigma

Background

Eating disorders (EDs) are a group of psychological illnesses that negatively affect eating behaviours and attitudes towards food, exercise, and the body, resulting in significant harm to overall health and well-being [1, 2]. Over one million individuals in Canada experience EDs and an estimated one in ten people will die because of their eating disorder (ED) [3, 4]. Despite these statistics, the current approach to publicly funded treatment in the Canadian context fails to meet the needs of many of those experiencing EDs.

Because healthcare is a provincial responsibility, publicly funded ED treatment differs in availability and modality across the country, creating a fragmented system of inequitable access and care [5]. Those in Canada requiring adult inpatient (IP) ED treatment (i.e., hospital-based, publicly funded care for those 18+needing medical stabilization) are particularly affected by this disconnection due to issues of structure, access, and efficacy. To our knowledge, no treatment standards for adult IP ED treatment have been developed and recognized on a national scale. Furthermore, access to adult IP ED treatment is scarce and unequal across the country, with several provinces having no adult IP ED programs at all [5]. This scarcity and geographic disparity inevitably contribute to growing wait times across the country, with some individuals in Canada waiting up to two years for life-saving ED treatment [6, 7]. Finally, the general efficacy of IP ED treatment remains contested, with studies suggesting high instances of patient drop-out and low rates of longterm ED symptom reduction [8–14].

One way to combat these issues is to work with individuals in Canada who have lived IP ED treatment experience to determine how they feel treatment programs and processes can be improved to better serve those in need of care [5, 11, 15]. While the research has begun to shift in the last 20 years, the valuable perspectives of those

with lived ED experience have historically been excluded from research dedicated to improving treatment outcomes due, in part, to stigma and ethical concerns [16]. Stigma at all levels has long been proven to impact how EDs are perceived, understood, experienced, and treated [17]. For example, social and professional stigma within the healthcare system suggests that ED patients are vain, manipulative, difficult to work with, and complex to treat, impacting the power dynamics between patient and caregiver and contributing to existing self-stigma that often accompanies an ED diagnosis [18–21]. Such stigma, combined with how quantitative data has often been privileged over qualitative research in clinical settings, likely contributes to the historic exclusion of the perspectives of those with lived ED experience in research [22].

In addition to stigma, there exists ethical concern about exposing those with lived ED experience to EDrelated topics for fear of facilitating relapse [16]. To mitigate potential risks and create an opportunity to explore the valuable perspectives of those with lived experience, protective methods can be put in place. This might include providing participants with recovery-oriented resources, facilitating open dialogue between researcher and participant, integrating supportive methods of data collection, and ensuring ongoing, informed consent [16]. In consideration of the potential risks and methods for risk mitigation, there has been an increase in qualitative, patient-centred ED research conducted over the past two decades. For example, two studies from the early 2000s explored adolescent experiences of IP ED treatment in the United Kingdom, bringing attention to peer relationships in treatment and the importance of a sense of identity and individuality [23, 24]. Another patient-centric study based in the United Kingdom was conducted in 2014 to explore patient perspectives of therapeutic alliance in IP ED treatment [25]. In 2019, two researchers sought to understand the general experiences of adolescents in IP ED treatment in China; a context in which the topic had previously been significantly under-researched [26]. More recently, a study based in the United States explored participants' experiences of ED treatment in the midst of the COVID-19 pandemic [27]. This form of inclusion, combined with input from healthcare professionals, has the potential to uncover novel findings that may aid in improving patient experiences and health outcomes, while also simultaneously addressing widespread stigma regarding EDs.

Though research centred around lived ED experience has increased in general, greater data must be collected across the globe in order to have meaningful and applicable impact in individual countries. Accordingly, such research specific to the Canadian context has historically been limited and in recognition of this, recent Canadian publications have urged researchers to seek further the perspectives of those with lived experience to aid in identifying areas for improvement of ED treatment [5, 11, 15]. This endorsement has resulted in several recent studies on the subject, though most research has been specific to youth and pediatric care. For example, Couturier et al. (2020) assembled a panel of stakeholders, including those with lived experience, to establish Canadian practice guidelines for the treatment of children and adolescents with eating disorders [28]. Also in 2020, a team of researchers in British Columbia, Canada sought to understand the treatment outcomes of youth (aged 16-24) in residential ED treatment via chart review in addition to participant interviews [29]. Finally, Coelho et al. (2022) conducted interviews with Canadian youth to explore their experiences with treatment, particularly noting the challenges they faced in accessing care [30]. While these studies are critical to improving care among a Canadian population that is significantly impacted by EDs, particularly since the beginning of the COVID-19 pandemic, input from Canadian adults is needed to develop a better system of care that meets the adult population's unique and diverse needs.

The purpose of this study, then, was to explore the perspectives of a sample of Canadian adults with lived experience of referral to and/or participation in adult IP ED treatment and identify participant-informed recommendations for the improvement of the processes associated with such treatment programs. This study not only responds to the call of recent Canadian publications to include those with lived experience in ED research, but it produces in addition a concrete list of detailed recommendations for the improvement of adult IP ED treatment in Canada that is entirely informed by those with lived experience [5, 11, 15].

Researcher positionality and reflexivity

I, Catherine Armour, undertook this work in partial fulfillment of my M.A. degree in Health Promotion from Dalhousie University and have lived experience with Canadian adult IP ED treatment. The methodology for this study was chosen intentionally to allow for this lived experience to be integrated into the research conduction rather than eliminated. My connection to the research topic aided in establishing mutual trust and respect between myself and the participants, allowing for a rich exploration of the research topic as perceived by participants. Seeking participants' feedback in developing the final list of recommendations ensured that they were reflective of the participants' experiences and not my own.

Methods

Philosophical worldview

A transformative philosophical worldview, defined by Mertens (2009) as an approach that aims to catalyze political and social change for the benefit of marginalized populations, guided this study [31]. Research that employs this worldview provides target populations, in this case, those in Canada with lived adult IP ED treatment experiences, the opportunity to have their voices heard and valued, meeting the purpose and rationale of this study [31, 32]. In alignment with this philosophy, those with lived ED experience were sought out to inform recommendations for change. This rejects existing discourse, upheld by existing literature that does not include lived experience, that healthcare professionals alone should inform treatment processes [32]. A transformative philosophical worldview is congruent with the methodology of this project in that it aims to bring attention to the experiences of marginalized populations and bring about change [31].

Theoretical framework

Feminist standpoint theory (FST) was employed for this research because of how patriarchy has shaped societal perceptions of EDs, who experiences them, and how they are treated [33, 34]. FST acknowledges that those who hold power over others in certain settings can shape and promote societal views on how these populations are perceived and therefore treated within society [35]. These perceptions (e.g., EDs are a women's issue, EDs are a result of vanity) are often inaccurate and perpetuated by stigma at the societal level [17, 36, 37]. FST theorizes that such ideologies can be shifted by seeking and valuing lived experience in research [35]. This theoretical framework informed the data collection process, ensuring participants a safe space to share their experiences, and the data analysis process (see Discussion), referring to feminist theories regarding gender, lived experience, and power in the context of IP ED treatment.

Research design

Given the sensitive nature of this study, a qualitative approach to data collection was regarded as the most appropriate study design to fulfill the purpose of the research. Conducting inductive qualitative research as an individual with lived experience ensured that data were collected respectfully and were reflective of the participants' experiences. The study followed hermeneutic phenomenology; a research design aimed at exploring lived experience to gain new understanding and knowledge on a phenomenon [38]. Hermeneutic phenomenology was most congruent with the transformative philosophical worldview, FST, and the overarching objective of the research [31, 35, 38]. Further, this approach appreciates, rather than prohibits, the connection between researcher and participant, acknowledging its value in the data collection and analysis processes [38, 39]. Building rapport with participants through our common experience was necessary to foster meaningful conversations and facilitate rich data collection. By employing hermeneutic phenomenology, explicitly stating my positionality, and engaging in peer review throughout the research process, I was able to conduct research that was reflective of the participants' experiences and perspectives without omitting my interpretations as a researcher with lived experience. Given my identity as someone with lived ED treatment experience, it is important to acknowledge my experience and its connection to this research.

Participants

Participants were recruited by sharing physical and electronic copies of a poster with ED support organizations across Canada; ten of which shared the poster with their networks. Potential participants were screened via email for their eligibility (see Table 1). They were also asked to share general information regarding their experience with adult IP ED treatment in Canada (i.e., age(s), location(s), year(s)) per maximum variation purposeful sampling which, in the context of this study, aimed to develop a participant sample that was diverse in their experience of the phenomenon of interest [40]. Participants were compensated \$15 for their participation in a virtual one-on-one interview and an additional \$5 for their participation in a virtual one-on-one follow-up meeting.

 Table 1
 Participant inclusion criteria

Inclusion Criteria

Lives in Canada

English fluency

Referred to an IP ED treatment program in Canada between the years 2012 and 2022 while aged 18 years or older

Not currently in IP ED treatment

Data collection

Qualitative data was collected via virtual, semi-structured, one-on-one, open-ended interviews and virtual, one-on-one follow-up meetings between the participant and me. Interviews and follow-up meetings were conducted virtually due to my geographic location at the time and the increased reach that virtual data collection allowed. Interviews lasted between 60 and 90 min, while follow-up meetings lasted between 20 and 45 min.

An interview guide was followed, however; the semistructured nature of the interviews allowed for flexibility to encourage exploration of each participant's perspective. The interviews generally followed three categories of questions: referral, transitions, and treatment. 'Referral' being the process associated with seeking and receiving a referral to an adult IP ED treatment program in Canada, 'transitions' meaning the processes associated with transition into and out of adult IP ED treatment program(s) in Canada, and 'treatment' referring to the processes associated with participating in adult IP ED treatment program(s) in Canada. Participants discussed elements within each category that they felt impacted their experience before identifying recommendations for how these elements might be improved to better support individuals in Canada.

Data analysis

Data were analyzed using interpretative phenomenological analysis (IPA) [41]. Interview transcripts were developed with assistance from a transcription software [42]. The transcripts were reviewed in-depth, and data were categorized. Recurring topics and recommendations were identified and were connected to related topics and recommendations that transpired in other participants' accounts. Common topics/recommendations identified in this process formed the draft list of recommendations. Eight of the eleven participants engaged in follow-up meetings where they shared their feedback on the draft list and prioritized the recommendations. This method of peer review was to ensure that my interpretations accurately reflected the participants' experiences and perspectives. The feedback informed the development of the final list of recommendations for the improvement of adult IP ED treatment in Canada. Ten categories of recommendations were identified; five of which are presented and discussed in this article due to their relevance to stigma and EDs.

Results

The following section will present the demographics of the participants and then define the participant-informed recommendations for the improvement of adult IP ED treatment in Canada. Recommendations are supported by the experiences of participants and excerpts from their virtual one-on-one interviews and follow-up meetings.

Demographics

The study sample was made up of eleven participants who met the inclusion criteria and represented the geographic diversity of lived Canadian adult IP ED treatment experience. Note that the demographic results only reflect what participants felt comfortable sharing. Participants were aged 22-37 years and represented ED diagnoses of anorexia nervosa (AN) (n=7), bulimia nervosa (BN) (n=1), and other specified feeding or eating disorder (OSFED) (n=1). Participants self-identified as non-binary (n=1), [cis-gender] female (n=9), and trans masculine non-binary (n=1). The study sample described their race/ethnicity as either Caucasian (n=10) or Moroccan Jewish (n=1) and their sexual orientation as queer (n=3), heterosexual (n=5), bisexual (n=2), or asexual (n=1). Study participants had lived adult IP ED treatment experience in Nova Scotia (n=1), Ontario (n=9), Alberta (n=2), and British Columbia (n=1). Several participants had been admitted to programs in several of the listed provinces. Demographic results have not been matched with individual participants to protect their identity.

Recommendation 1: interim support

As made evident through participant experiences, the lack of accessible interim support (i.e., treatment during the waiting period between referral and admission to adult IP ED treatment) combined with rigid eligibility criteria surrounding publicly funded treatment had significant consequences on the health of study participants. In recognition of this, our first recommendation is that support must be available in the interim to prevent patient health deterioration and improve overall well-being. Interim support programs and services, (e.g., meal support, psychotherapy, dietetics, medical monitoring) must be available to all incoming patients and must be made accessible. Patients' geographical location and non-ED commitments (i.e., family, education, employment) must be accommodated when designing accessible interim support programs.

Upon referral to treatment, participants recalled feeling guilty and shameful about their impending admission to adult IP ED treatment. These feelings were perceived to be influenced by internalized stigma due to a lack of resources in adult IP ED treatment in Canada. Participant Two described these feelings saying:

That I had to go into a hospital - that made absolutely no sense to me ... I had this overwhelming sense of guilt about taking up a bed ... I believed ... I didn't need that level of care and that by taking

that level of care, I would be doing something really, really wrong by taking it away from someone who might need it more.

In addition to feelings of guilt and shame, the 'not sick enough' mindset (i.e., feeling unworthy of care) was common among participants. Participant One eluded to this mindset, saying: "[I felt] a lot of self-judgment [upon referral] ...the whole 'I'm not sick enough' mindset, and 'I'm undeserving of care' and 'people need it more than I do'", describing the self-judgment that accompanied referral to adult IP ED treatment.

Long wait times were also of concern among the study sample. Participants waited up to 15 months between referral and admission. During this waiting period, every participant reported a decline in their physical and mental health. This decline resulted in medical emergencies in some participants, including Participant Three, who said: "I depleted so badly that I ended up having a heart attack and a stroke". Participant Two had similar experiences, stating: "[I had] to go to emergency multiple times while waiting for treatment". The commonality of long wait times and medical emergencies in the interim was further underscored by another comment from Participant Three: "I can honestly say I've known people personally who have died on that waitlist". Waiting for admission to IP ED treatment was agonizing for participants, not only due to the fleetingness of motivation to recover but also because of the steep decline in health that they experienced while waiting.

Access to interim support was uncommon for study participants, with only one participant receiving specialized publicly funded outpatient treatment prior to their IP admission. Other participants sought private support in the interim, while most study participants were left with no interim support and were instead left to rely on unspecialized care via local health clinics and emergency room visits. Eligibility criteria (i.e., minimum/maximum body mass index (BMI) requirements) for adult IP ED treatment restricted many study participants from enrolling in publicly funded outpatient programs. Participant Eight described their experience of awaiting IP ED treatment admission:

I felt like the referral process...almost incentivized you to get worse...for so long, I was in a place where I was not quite healthy enough for outpatient services, but not quite sick enough for IP services...it felt so much easier to get worse and get care through IP than get better and get help through outpatient.

Private treatment options allowed more flexibility in eligibility criteria, yet this form of support is costly and also proved to be unreliable in the interim. Participant Two, said, for example: "I was seeing a therapist [and a] dietician and we were not making progress and they were saying like, we can't work with you anymore", describing their experience of having a private care team while awaiting admission to IP ED treatment, however; after a lack of progress, this form of support was terminated.

Recommendation 2: dignified treatment

Participants expressed a lack of trust and a culture of disrespect in their experiences of adult IP ED treatment, which left them feeling undignified. Because of this, our second recommendation is that treatment must be respectful of patients' dignity. Maintaining all facets of a patient's life should be prioritized while in treatment. The program should collaborate with a patient's support system to determine and implement an appropriate treatment plan to allow this. All processes should be conducted with compassion, empathy, and understanding for the patient. Patients should not be stripped of their autonomy and independence. These actions require healthcare providers to implement principles of patient-centered care.

Upon admission to treatment, participants felt a power imbalance between the patients and the healthcare providers, leaving patients feeling unwelcome, powerless, and even unsafe. While certainly not the intention of all healthcare providers, this was a common experience among study participants. Participant Four reflected on their first days in treatment, recalling a 'tough love approach': "[It was] probably what I needed at the time, but it felt very jarring, and I could see a lot of people not doing well with that". Participant One supported this sentiment, saying: "I think there's room for improvement to really meet the patient's needs rather than being a drill sergeant". This power imbalance and culture of disrespect led some participants, including Participant Nine, to feel less than human: "I generally just felt like an animal". This statement summarizes participants' common perception of treatment as being, at times, inhumane.

In addition to the culture of disrespect and power imbalance, participants noticed a lack of trust from healthcare providers. Participants felt that the patient perspective was often disregarded and not believed. Participant Seven described this phenomenon by referencing an interaction with a healthcare provider: "They kind of looked at me like, 'Well, you're the crazy one"." Participant Three also recalled an instance during which they were not trusted by healthcare providers in an adult IP ED treatment program. Prior to a routine weigh-in, the participant was not allowed supervised use of the bathroom, resulting in an elevated weight on the scale. This participant was accused of intentional water-loading and was discharged from the program without an opportunity for explanation. Participant Three described the situation:

"The weight on the scale was so significantly high, because I had so much fluid in me...[that] I got discharged". Experiencing this distrust from healthcare providers discouraged participants in their recovery journey, making them feel as though they were alone in their hardships.

Administrative discharge (i.e., getting kicked out of a program prematurely) was a common experience among participants. Those who were administratively discharged felt there was a lack of discharge planning, as in Participant One's experience: "I was left with zero follow-up...if you're not successful, you don't get follow-up at all". This meant that not only were patients being discharged prematurely, but they were also being discharged without support. Participant Seven was discharged in a concerningly dangerous physical state: "They discharged me at like, I think I was maybe 68 pounds". Being discharged without discharge planning at such a low weight could have had fatal consequences.

Contrarily, participants alluded to several instances of respect and compassion from staff, and these moments were described fondly during interviews. Participant Three recalled a particular time when their doctor comforted them through a difficult time in treatment: "[The doctor] came up to my bed that night ... as he was getting ready to leave and he ... pulled up the chair right beside my bed and we played cards and he says, I don't care how long this takes. When you're ready. You can talk to me.". This moment was described as a "breakthrough" for this participant. Participant Five agreed that "building rapport with staff" was important in recovery and Participant Ten reflected on dignity in treatment, saying, "It wasn't necessarily that they had to be specialists in eating disorders. They ... just needed to be human with me...making you feel like you're on more equal ground with them can really make a big difference". These positive reflections further underscore the value of and the need for dignified ED care.

Recommendation 3: individualized care

Participants felt that the approach to adult IP ED treatment in Canada was too generalized and ignorant of how social determinants of health (SDOH) influence individual patients' experiences of EDs, recovery, and treatment. In response, our third recommendation is that all processes must be conducted in recognition of the fact that every person is different, and every ED experience is different. Referral processes should be more subjective and holistic; factors beyond just weight and BMI must be considered when evaluating the need for treatment. SDOH and the unique factors that impact a patient's experience of an ED, treatment, and recovery must be considered throughout all treatment processes. Future ED research in Canada should be focused particularly on improving the health outcomes of marginalized populations.

Participants noted a 'one size fits all' approach to treatment taken by programs, and how ineffective it can be. Participant Three summarized this common notion, saying: "Maybe the ... one size fits all treatment isn't actually fitting all". The ineffectiveness of this approach was even more evident in the experiences of participants who were diverse (i.e., in comparison to other study participants) in culture, gender, or housing. As a Moroccan Jew, Participant Eight reflected on their experience seeking spiritual support while in treatment: "They told me that they'd connect me to the spiritual care section of the hospital. Every single person who worked in the spiritual care unit was a white Christian lady ... [who asked me to] teach her about Judaism." Not only did this lack of individualized care leave Participant Eight with insufficient support, but it also left them feeling alone in their experience: "It just made me feel like eating disorder treatment is only for like, white cisgender Christian women". This marginalization left Participant Eight unsupported and with reduced hope for recovery.

Participants who identified as transgender and gender non-binary voiced additional concern for the lack of consideration given to the unique experiences and challenges of gender-diverse populations in IP ED treatment. Participant Eight reflected on their experience, saying: "Weight gain was really hard for me, because of like gender and stuff ... and they were like, you just need to accept your secondary sex characteristics ...we're not a gender therapy, we're an eating disorder therapy.". Participant Eleven expressed similar sentiments: "Being someone who is ... trans non-binary, so much of eating disorder treatment ends up catering to cis-women and I think that, like, there's a huge gap in education". These participants felt that gender affirmative care and attention to the roles of gender dysphoria in ED development and recovery were missing from their treatment experience. Instead, therapeutic approaches were largely built on hetero-cisnormativity, addressing topics like body image and societal expectations of femininity which held little significance in their own experiences. The perspectives of Participants Eight and Eleven provide an example of the alienation and marginalization that can arise when individualized care, particularly care that is culturally competent and reflective of one's gender identity, is not prioritized.

Because of the lack of individualization in adult IP ED treatment, housing was another SDOH not considered while in care which caused an issue for Participant Ten. This participant refrained from sharing their experience with housing instability while in treatment due to the outcome of a co-patient: "They had kicked out someone else ... because she had mentioned that she didn't have secure housing, basically ... so I wasn't going to be like, 'Hi, I don't have housing either". Concealing this concern created an opportunity for the participant to feel isolated

and it reaffirmed the existing power imbalance between healthcare provider and patient.

Contrarily, there were several times when participants felt that their individual needs were considered and met in treatment. For example, when participants' individual medical needs were met while in treatment or when participants could collaborate with a staff member on an individualized therapy plan, daily schedule, or discharge plan, participants felt supported. Participant Two said, "They were really thorough with the medical testing ... that is the way I felt most supported". Another participant reflected positively on a time when their admission day was moved so that they could spend a religious holiday at home with family. These experiences emphasize the benefit of, and further stress the need for, care that is tailored to the needs of each individual patient.

Recommendation 4: resources

Many areas for improvement identified by participants can only be remedied with increased resources allotted to Canadian adult IP ED treatment. In consideration of this, our fourth recommendation is to allot sufficient resources (i.e., funding, staff, training, beds, programming) to this form of care to increase accessibility of treatment and improve population health outcomes. Participants' experiences with the lack of interim support, long wait times, disrespectful and untrusting healthcare providers, and generalized treatment approaches all support the need for the implementation of this recommendation.

In addition, there are several participant experiences and excerpts that relate directly to the need for increased resources. Participants recognized that many of their negative experiences with treatment were simply a product of insufficient funding and resources. Participant Eight shared: "The psychiatric system...for me felt like they weren't there to offer support for my health, they were more there to keep it [away] from people and I know that's like [an] issue with funding". Participant One expressed a similar sentiment: "I think they genuinely do what they can and what their limited funding and resources allow for, but the funding [and] resources aren't adequate". It was common for participants to share that they felt poorly that many of their reflections on treatment were negative. Participants were sympathetic towards healthcare providers within the treatment programs as they saw first-hand just how stretched the resources were. During Participant Two's follow-up meeting, they clearly stated their feelings towards the fifth recommendation, saying: "The bottom line is that there needs to be so much more resources for [the implementation of the identified recommendations] to happen". Participants felt strongly that increased resources must be prioritized.

Recommendation 5: stigma

Given the negative impact of various levels of stigma (i.e., internalized, societal, institutional) evident in the participants' experiences, our fifth recommendation, alongside implementing the aforementioned recommendations, is to apply the following methods for active stigma reduction. Educational curriculums must be established and/or updated to generate accurate and widely understood depictions of EDs, who experiences them, and how they can be treated. People with lived ED experience must be included at the forefront in the development of policies, guidelines, and practices related to EDs. Finally, referral, transitional, treatment, and hiring processes must better reflect the diversity of populations impacted by EDs.

Stigma played a direct, negative role in participants' experiences of how they felt being referred to IP ED treatment, the way they were perceived and treated by healthcare providers, and the generalized approach to IP ED treatment as discussed in previous sections. As with the fourth recommendation, several participant experiences and excerpts have been identified to further support this recommendation. Participants acknowledged the stigma associated with experiencing an ED at a normal or high weight, and how this stigma impacts one's access to referral and treatment. Participant Six said: "If I, today, just stayed in this body [as someone who lives] in a relatively thin body...I would be dealing with people going, 'Oh, but you're healthy, what do you mean you're sick?'...That would add a whole other layer to seeking support and also being taken seriously [in treatment]". Participant Three experienced this stigma first-hand as someone who was admitted to treatment several times at varying weights: "My weight was fluctuating every admission...[and] how I got treated and their approach to me completely changed". Many participants already felt guilty about being referred to treatment or 'not sick enough' to deserve treatment, and experiences like that of Participant Three contribute to the stigma that perpetuates these feelings. Stigma also affected the relationships between participants and healthcare providers. Participant One eluded to this stigma, saying: "I think even eating disorder professionals carry somewhat of a stigma with them [toward patients]". This stigma likely contributed to the culture of disrespect and distrust that was commonly experienced by participants in treatment.

Discussion

This study explored perspectives on improving adult IP ED treatment of individuals in Canada with lived experience, identifying recommendations categorized as interim support, dignified treatment, individualized care, resources, and stigma. The discussion will place the study sample, their lived experiences, and the identified recommendations within the context of the existing literature

before acknowledging the strengths and limitations of the study, and suggestions for future research.

The demographics of the study sample reflect trends in relevant literature regarding who can typically access ED diagnosis, referral, and treatment. Many adult IP ED treatment programs in Canada only treat AN and BN, and the average onset for these diagnoses is between the ages of 18 and 21 [5, 43, 44]. These data can explain the lack of diversity in diagnoses in this study's sample, as well as the ages of participants being relatively young. Further, people of colour and Indigenous populations experience unique and disproportionate barriers to ED diagnosis, referral, and treatment, hence the lack of racial/ethnic diversity in the study sample [37, 45]. Finally, adult IP ED treatment exists in only seven provinces, therefore; lived experience from programs in Manitoba, Quebec, and Newfoundland was missing from this study.

According to current data, individuals in Canada are waiting up to 2 years to be admitted to treatment which is dangerous considering the high mortality rate of EDs [3, 6, 7]. The inaccessibility of this sort of treatment is largely due to the scarcity of adult IP ED treatment programs in Canada and spots within those programs [5]. These data mirror the experiences of the study sample who waited, on average, six months for admission following referral, during which time, feelings of guilt, shame, and not being 'sick enough' were common. Existing literature suggests that shame is commonly associated with diagnoses of AN and BN, and skewed self-perception of the illness' severity contributes to the challenges associated with seeking and accepting ED treatment [46-48]. More recently, researchers have explored the role that diagnostic criteria for OSFED, particularly for Atypical AN, have in contributing to stigma and the 'not sick enough' mindset described above. Atypical AN is diagnostically the same as AN, however; the weight of individuals with Atypical AN falls within a 'normal' range. Research has suggested that this distinction can facilitate self-stigma among those of all ED experiences and diagnoses and even translates to under-recognition from clinicians [49, 50]. This research challenges the long-standing use of weight and BMI as indicators of ED severity and could aid in alleviating the widespread stigma that impacts the health of individuals with EDs as it did in this particular study.

Lack of interim support contributed most significantly to participants' health decline while awaiting admission to IP ED treatment. According to MacNeil et al. (2016), publicly funded adult outpatient ED treatment programs across Canada typically require patients to have a BMI of 16 kg/m² [51]. Patients with a lower BMI are referred to IP ED treatment as they require medical stabilization. This leaves patients awaiting admission to IP

ED treatment without interim support, as was the case for the majority of this study's participants. MacNeil et al. (2016) studied an Ontarian woman awaiting IP ED treatment, enrolled her in an outpatient program, and reported increased body weight, increased satisfaction with life, and decreased depressive and anxiety symptoms in the woman [51]. This study suggests that outpatient treatment can benefit those awaiting admission to adult IP ED treatment, despite not meeting outpatient BMI criteria [51]. These findings in addition to the experiences of participants contributed to the establishment of *Recommendation 1: Interim Support*, with accessibility of care being paramount to its implementation.

The culture of distrust and disrespect from healthcare providers noticed by participants could be linked to the pathology of EDs and the techniques employed in IP ED treatment. EDs often develop and/or are sustained in an attempt to gain control over one's life when other elements feel out of control. Having 'control' in the form of an ED can offer a sense of security [52]. In the experiences of the study's participants, this 'control' was taken over entirely by healthcare providers, creating a clear power shift upon entry into treatment, and resulting in feelings of resentment, fear, and frustration among patients. The role of power in healthcare has long been a prominent subject within feminist literature. Take for instance an ethnographic study of an IP ED treatment facility based in the United States in the early 2000s. Employing a feminist lens, Gremillion (2003) critically analyzed the facility's approach to care, noting the impacts of power, control, and surveillance on relationships and patient outcomes. Gremillion (2003) compared healthcare providers to 'parents' within a patient's 'therapeutic family'; the psychologists (usually male) exercising more abrasive power over other staff and patients, and nurses (usually female) taking a more nurturing, though still powerful, approach with patients. The study suggested that the very socio-cultural aspects that contribute to ED development may be reinforced by the gendered and hierarchal power dynamics within the clinical setting [53].

In 2022, a group of scholars produced two papers inviting productive conversations about feminism and the spectrum of EDs. The second paper involved reflection on how feminist theories can practically inform care. In relation to power and control, LaMarre et al. (2022) emphasized the importance of collaborative care and suggested that power-sharing could aid in gaining power equilibrium between healthcare provider and patient, and ultimately contribute to improved patient outcomes [54, 55]. This suggestion is further supported by additional existing literature which suggests that ED patients respond best when healthcare providers are knowledgeable about EDs, respectful of patients' need

for autonomy, and collaborative in ways that encourage autonomy, agency, and empowerment of patients [18].

Research exploring perceptions of ED patients suggests that some healthcare providers stigmatize those with EDs as vain, manipulative, non-trustworthy, and challenging to work with [19, 56]. A study from the United Kingdom suggests that patient-centred care with a focus on fostering a culture of hope and respect shows promise in treating those with EDs [57]. In consideration of the importance of collaboration, compassion, autonomy, and empowerment in positive health outcomes and the general experiences of adult IP ED treatment in Canada, *Recommendation 2: Dignified Care* was proposed with an emphasis on integrating collaborative, patient-centred care.

The Public Health Agency of Canada (2016) considers SDOH (e.g., gender, race, housing, income, religion, culture) to be factors that affect the health of populations [58]. Despite these SDOH and their impact on patients' experiences of EDs and ED treatment, Canadian adult IP ED treatment programs, as per study participants' experiences, typically employ a 'one size fits all' approach. This approach remained unchanged regardless of patients' diverse traits, experiences, and needs. The reason for this approach may be linked to the widely understood and fiercely inaccurate SWAG stereotype, which assumes only skinny, white, affluent girls can develop an ED [37].

The experiences of participants who did not fit the SWAG stereotype further support the linkage between the SWAG stereotype and the 'one size fits all' approach to treatment. For example, Participant Eight was denied the privilege of spiritual support while in treatment due to their identity as a Moroccan Jew despite evidence that suggests spiritual health to be integral and powerful in the success of some patients' ED recovery [59]. Research has found that transgender and gender non-binary populations experience EDs and ED recovery in unique ways; changes to body size and shape facilitate gender dysphoria and increase the complexity of recovery, yet gender was excluded from treatment entirely, creating challenges for Participants Eight and Eleven [60]. Participant Ten, who experienced housing instability while in treatment, felt unsupported due to the stigma that suggests housing instability cannot coexist with maintaining recovery [61]. Feminist scholars generally criticize the medical model of EDs, which considers the illness a biological disorder, and favour instead sociocultural models which argue that EDs are inherently embedded within an individual's social surroundings and cultural environment, and therefore cannot be prevented, diagnosed, or treated effectively without acknowledging such context [62, 63]. In consideration of this, participants identified Recommendation 3: Individualized Care in hopes that adult IP ED treatment programs and associated processes can adapt to meet the

needs of individuals in Canada requiring care rather than expecting the inverse.

The aforementioned recommendations require increased resources to come to fruition. To reiterate previous sentiments, adult IP ED treatment is not available in all provinces and territories in Canada [5]. Across Canada, hospitals are understaffed, and healthcare providers are overworked, leading to burn-out [64, 65]. Canadian medical students participate in only three to five hours, on average, of ED-specific training and a recent study estimates that health and social costs of ED treatment among youth have increased by at least 21% during the COVID-19 pandemic [66, 67]. Increased resources are required to remedy these shortfalls and implement all recommendations identified in this study, hence the development of Recommendation 4: Resources.

The need for increased resources spans beyond just the scope of EDs in the Canadian context; greater resources, particularly funding, are in demand across the country for mental health and substance-use health challenges of all kinds. A 2018 Statistics Canada survey found that, of the 5.3 million Canadians in need of mental health care, nearly half of them felt that their needs were partially or fully unmet due to barriers including lack of access and cost [68]. Despite the rising costs associated with ED treatment since the pandemic; an estimated increase of at least 21% for Canadian youth, funding for ED prevention, treatment, and research still trails behind that allotted to other psychiatric illnesses [67, 69]. For example, where approximately \$50.17 per person affected is spent annually on schizophrenia research, only \$0.70 per person affected is spent on ED research [70]. These data emphasize the need for increased resources across all psychiatric illnesses, with a particular focus on funding for EDs.

Implementing the above recommendations will aid in addressing the stigma evident in participants' experiences, but more direct action is needed as well. As recommended by Canadian ED publications, further research is required to explore the diverse lived experiences of individuals in Canada with EDs, increase awareness on EDs, and catalyze change [5, 11, 15]. An international study of 24 countries found that 73% of the 512 ED-specialized healthcare providers were white, 89.6% were women, and 84.1% were heterosexual [71]. This fact alone once again reinforces the SWAG stereotype and brings attention to the need for ED treatment and the staff who administer it to reflect the diversity of those requiring care. It is due to the immense impact that stigma has on those with EDs in Canada and their experiences accessing and navigating recovery that Recommendation 5: Stigma was identified in this study.

Strengths, limitations, and suggestions for future research

This study is qualitative in design, specific to adults in Canada, and explores lived experiences from within the last decade, addressing gaps in the existing Canadian literature where studies are often quantitative, youth-focused, or outdated. Additionally, it achieves suggestions put forth by several Canadian publications to include lived experience at the forefront when developing methods for the improvement of Canada's approach to EDs. This research is significant in that it provides detailed recommendations, informed by lived ED treatment experience and supported by existing literature, that have the power to positively impact future generations.

This study is only reflective of the lived experiences of eleven participants who expressed interest in the study, met inclusion criteria, and were selected to participate according to the sampling technique. Countless perspectives are missing from the development of these recommendations and these perspectives must be sought, included, and valued in future studies. In particular, the perspectives of people of colour, Indigenous communities, transgender and gender non-binary populations, men and especially gay men, people with disabilities, people in bigger bodies, older populations, and people with less-commonly treated ED diagnoses must be sought, included, and valued. Ideally, this research should be conducted by researchers identifying with the target population(s).

Conclusions

This study explored lived experience as a catalyst for improvement to current adult IP ED treatment programs and processes in Canada; an exploration necessitated by the need for increased first-voice perspectives in the existing Canadian literature. Participants reflected on their experiences of referral, transitions, and treatment itself and identified recommendations for the improvement of adult IP ED programs and processes. It was found through working collaboratively with people with lived ED experience that interim support, individualized care, dignified treatment, increased resources, and stigma-reducing strategies must be prioritized and implemented with haste to improve the health outcomes of those in Canada and the overall experience of IP ED patients while also addressing individual and societal levels of stigma contributing to the negative experiences of people with EDs. It is evident that people with lived ED experience have considerable insight into how treatment approaches can be improved, and how stigma can be reduced; therefore, their perspectives and opinions must be esteemed in future research and program/policy development. In particular, the lived ED experience of people not fitting within the SWAG stereotype must be explored to develop an approach to care and stigma

reduction that reflects the diversity of those impacted by EDs.

Abbreviations

EDs Eating disorders ED Eating disorder IP Inpatient

SWAG Skinny, white, affluent girls SDOH Social determinants of health

BMI Body mass index

FST Feminist standpoint theory

IPA Interpretative phenomenological analysis

AN Anorexia nervosa BN Bulimia nervosa

OSFED Other specified feeding and eating disorder

Acknowledgements

I would like to extend my gratitude for the vulnerability and openness of the study participants. I would also like to thank the eating disorder support organizations across Canada that shared the recruitment poster, and my Master's supervisory committee who provided guidance in this research.

Author contributions

Miss Armour conceived the study, conducted the data collection and analysis, and wrote the manuscript draft. Dr. Gahagan served as the co-supervisor for Miss Armour's MA Health Promotion thesis, and Dr. Feicht served as a committee member; they provided mentorship, guidance, and supervision for the study, thesis development and defense, and manuscript preparation and revision.

Funding

This project was funded by the Canadian Institutes of Health Research (CIHR) Canadian Graduate Scholarships – Master's (CGS-M) program.

Data availability

To maintain confidentiality of participants, the data collected and analyzed for this study are not publicly available.

Declarations

Ethics approval and consent to participate

This study was approved by Dalhousie University's Research Ethics Board. The participants were provided written information on the study and all signed informed consent documents prior to their participation. Participants were informed of their ability to terminate their participation at any time. Identifiable information was removed to protect participants.

Consent for publication

Not applicable.

Competing interests

The authors declare no competing interests.

Additional information

Not applicable.

Author details

¹Dalhousie University, Halifax, NS, Canada ²Mount Saint Vincent University, Halifax, NS, Canada

Received: 28 February 2024 / Accepted: 27 September 2024 Published online: 22 October 2024

References

- American Psychiatric Association. Diagnostic and statistical Manual of Mental disorders – Fifth Edition (DSM-5). American Psychiatric Association; 2013.
- National Eating Disorder Information Centre. Types of Eating Disorders. 2021. https://nedic.ca/eating-disorders-treatment/ Accessed 1 Feb 2024.

- Arcelus J, Mitchell AJ, Wales J, Nielsen S. Mortality rates in patients with anorexia nervosa and other eating disorders: a meta-analysis of 36 studies. Arch Gen Psychiatry. 2011;68(7):724–31. https://doi.org/10.1001/ archgenpsychiatry.2011.74.
- Galmiche M, Déchelotte P, Lambert G, Tavolacci MP. Prevalence of eating disorders over the 2000–2018 period: a systematic literature review. Am J Clin Nutr. 2019;109(5):1402–13. https://doi.org/10.1093/ajcn/nqy342.
- Canadian Eating Disorders Alliance. The Canadian Eating Disorders Strategy: 2019–2029. 2019. https://nedic.ca/media/uploaded/Canadian_Eating_Disorders_Strategy_2019-2029_-dps_-_ENG.pdf Accessed 18 Dec 2023.
- CBC/Radio Canada. Wait times for eating disorder treatment in Canada grow during the pandemic. CBC News. 2022. https://www.cbc.ca/news/health/ wait-times-for-eating-disorder-treatment-in-canada-grow-during-the-pandemic-1.6533635 Accessed 29 Dec 2023.
- Keski-Rahkonen A, Mustelin L. Epidemiology of eating disorders in Europe: prevalence, incidence, comorbidity, course, consequences, and risk factors. Curr Opin Psychiatry. 2016;29(6):340–5. https://doi.org/10.1097/ YCO.000000000000278.
- Fassino S, Piero A, Tomba E, Abbate-Daga G. Factors associated with dropout from treatment for eating disorders: a comprehensive literature review. BMC Psychiatry. 2009;9(67). https://doi.org/10.1186/1471-244X-9-67.
- Fichter MM, Quadflieg N, Hedlund S. Long-term course of binge eating disorder and bulimia nervosa: relevance for nosology and diagnostic criteria. Int J Eat Disord. 2008;41(7):577–86. https://doi.org/10.1002/eat.20539.
- Fichter MM, Quadflieg N, Crosby RD, Koch S. Long-term outcome of anorexia nervosa: results from a large inpatient longitudinal study. Int J Eat Disord. 2017;50(9):1018–30. https://doi.org/10.1002/eat.22736.
- LeBlanc H. Eating Disorders Among Girls and Women in Canada: Report of the Standing Committee on the Status of Women. 2014. https://novascotia. cmha.ca/wp-content/uploads/2018/01/Eating-Disorders-Among-Girls-and-Women-in-Canada-pdf Accessed 10 Jan 2024.
- Masson PC, Perlman CM, Ross SA, Gates AL. Premature termination of treatment in an Inpatient Eating Disorder Programme. Eur Eat Disorders Rev. 2007;15:275–82. https://doi.org/10.1002/rev.762.
- Quadflieg N, Fichter MM. Long-term outcome of inpatients with bulimia nervosa-results from the Christina Barz Study. Int J Eat Disord. 2019;52(7):834– 45. https://doi.org/10.1002/eat.23084.
- Walker J, Vibert S, Berthoz S, Huas C, Hubert T, Godard N. Dropout from Inpatient Treatment for Anorexia Nervosa: critical review of the literature. Int J Eat Disord. 2009;42(7):636–47. https://doi.org/10.1002/eat.20609.
- Obeid N, McVey G, Seale E, Preston W, Norris ML. Correlating research priorities for anorexia nervosa: the Canadian Eating Disorder Priority setting. Int J Eat Disord. 2020;53(5):662–72. https://doi.org/10.1002/eat.23234.
- Musić S, Elwyn R, Fountas G, Gnatt I, Jenkins ZM, Malcolm A, Miles S, Neill E, Simpson T, Yolland C, Phillipou A. Valuing the voice of lived experience of eating disorders in the research process: benefits and considerations. Australian New Z J Psychiatry. 2022;56(3):216–8. https://doi.org/10.1177/000486742199879.
- Brelet L, Flaudias V, Désert M, Guillaume S, Llorca P-M, Boirie Y. Stigmatization toward people with Anorexia Nervosa, Bulimia Nervosa, and binge eating disorder: a scoping review. Nutrients. 2021;13(8):2834. https://doi.org/10.3390/ nul3082834
- Johns G, Taylor B, John A, Tan J. Current eating disorder healthcare services the perspectives and experiences of individuals with eating disorders, their families and health professionals: systematic review and thematic synthesis. BJPsych Open. 2019;5(4):e59. https://doi.org/10.1192/bjo.2019.48.
- Thompson-Brenner H, Satir DA, Franko DL, Herzog DB. Clinician reactions to patients with eating disorders: a review of the literature. Psychiatric Serv. 2012;63(1):73–8. https://doi.org/10.1176/appi.ps.201100050.
- Johns G, Taylor B, John A, Tan J. Current eating disorder healthcare services the perspectives and experiences of individuals with eating disorders, their families and health professionals: systematic review and thematic synthesis. BJPsych Open. 2019;5(4). https://doi.org/10.1192/bjo.2019.48.
- Bannatyne A, Stapleton P. Eating disorder patient experiences of volitional Stigma within the Healthcare System and views on Biogenetic Framing: a qualitative perspective. Australian Psychol. 2016;53(4):325–38. https://doi. org/10.1111/ap.12171.
- Al-Busaidi ZQ. Qualitative research and its uses in health care. Sultan Qaboos Univ Med J. 2008;8(1):11–9. PMID: 21654952; PMCID: PMC3087733.
- Colton A, Pistrang N. Adolescents' experiences of inpatient treatment for anorexia nervosa. Eur Eat Disorders Rev. 2004;12(5):307–16. https://doi. org/10.1002/erv.587.

- Offord A, Turner H, Cooper M. Adolescent inpatient treatment for anorexia nervosa: a qualitative study exploring young adults' retrospective views of treatment and discharge. Eur Eat Disorders Rev. 2006;14(6):377–87. https://doi.org/10.1002/erv.687.
- Sly R, Morgan JF, Mountford VA, Sawer F, Evans C, Lacey JH. Rules of Engagement: qualitative experiences of Therapeutic Alliance when receiving In-Patient treatment for Anorexia Nervosa. Eat Disord. 2014;22(3):233–43. https://doi.org/10.1080/10640266.2013.867742.
- Wu Y, Harrison A. Our daily life was mainly comprised of eating and sitting: a
 qualitative analysis of adolescents' experiences of inpatient eating disorder
 treatment in China. J Eat Disorders. 2019;7(1). https://doi.org/10.1186/
 s40337-019-0250-6.
- Goode RW, Godoy SM, Wolfe H, Olson K, Agbozo B, Mueller A, et al. Perceptions and experiences with eating disorder treatment in the first year of COVID -19: a longitudinal qualitative analysis. Int J Eat Disord. 2022;56(1):247–56. https://doi.org/10.1002/eat.23888.
- Couturier J, Isserlin L, Norris M, Spettigue W, Brouwers M, Kimber M, et al. Canadian practice guidelines for the treatment of children and adolescents with eating disorders. J Eat Disorders. 2020;8(1). https://doi.org/10.1186/ s40337-020-0277-8.
- Williams KD, O'Reilly C, Coelho JS. Residential treatment for eating disorders in a Canadian treatment centre: clinical characteristics and treatment experiences of residents. Can J Behav Sci / Revue canadienne des Sci du Comportement. 2020;52(1):57–66. https://doi.org/10.1037/cbs0000143.
- Coelho JS, Suen J, Marshall S, Zaid-Alkailani H, Geller J, Lam P. Treatment experiences of male and female youths with eating disorders. Br J Clin Psychol. 2022;61(4):1119–33. https://doi.org/10.1111/bjc.12384.
- 31. Mertens DM. Transformative research and evaluation. New York: Guilford Press: 2009.
- van Janse MG. Transformative eating disorder research: qualitative research considerations. J Eat Disorders. 2021;9(1). https://doi.org/10.1186/ s40337-021-00428-2.
- McBride HL, Kwee JL. Inside and Out: how western patriarchal cultural contexts shape women's relationships with their bodies. In: Stoyanov D, Fulford B, Stanghellini G, Van Staden W, Wong MT, editors. International perspectives in values-based Mental Health Practice. Cham: Springer; 2021. https://doi.org/10.1007/978-3-030-.
- Nash CJ, Patriarchy. Int Encyclopedia Hum Geogr. 2020;43–7. https://doi. org/10.1016/B978-0-08-102295-5.10206-9.
- Henwood K, Pidgeon N. Remaking the link: qualitative Research and Feminist Standpoint Theory. Feminism Psychol. 1995;5(1):7–30. https://doi. org/10.1177/0959353595051003.
- Holmes S. Blindness to the obvious? Treatment experiences and feminist approaches to eating disorders. Feminism Psychol. 2016;26(4):464–86. https://doi.org/10.1177/0959353516654503.
- Sonneville KR, Lipson SK. Disparities in eating disorder diagnosis and treatment according to weight statues, race/ethnicity, socioeconomic background, and sex among college students. Int J Eat Disord. 2018;51(6):518–26. https://doi.org/10.1002/eat.22846.
- McManus Holroyd AE. Interpretive hermeneutic phenomenology: clarifying understanding. Indo-Pacific J Phenomenology. 2007;7(2). https://doi. org/10.10520/AJA14457377_58.
- Neubauer BE, Witkop CT, Varpio L. How phenomenology can help us learn from the experiences of others. Perspect Med Educ. 2019;8:90–7. https://doi. org/10.1007/s40037-019-0509-2.
- Patton MQ. Qualitative research and evaluation methods. 3rd ed. Thousand Oaks, CA: Sage; 2002.
- Reid K, Flowers P, Larkin M. Exploring lived experience. Psychol. 2005;18(1):20–3.
- Otter.ai. Al meeting note Taker & Real-time Al transcription. Otter.ai Al Meeting Note Taker & Real-time Al Transcription. 2024. https://otter.ai/ Accessed 30 Jan 2024.
- Volpe U, Tortorella A, Manchia M, Monteleone AM, Albert U, Monteleone P. Eating disorders: what age at onset? Psychiatry Res. 2016;30(238):225–7. https://doi.org/10.1016/j.psychres.2016.02.048.
- Ward ZJ, Rodriguez P, Wright DR, Austin SB, Long MW. Estimation of eating disorders Prevalence by Age and associations with Mortality in a simulated nationally Representative US Cohort. JAMA Netw Open. 2019;2(10):e1912925. https://doi.org/10.1001/jamanetworkopen.2019.12925.
- 45. Mikhail ME, Klump KL. A virtual issue highlighting eating disorders in people of black/African and indigenous heritage. Int J Eat Disord. 2020;54(3):459–67. https://doi.org/10.1002/eat.23402.

- Anderson LM, Hall LMJ, Crosby RD, Crow SJ, Berg KC, Durkin NE, Engel SG, Peterson CB. Feeling fat, disgust, guilt, and shame: preliminary evaluation of a mediation model of binge-eating in adults with higher-weight bodies. Body Image. 2022;42:32–42. https://doi.org/10.1016/j.bodyim.2002.05.008.
- Blythin SPM, Nicholson HL, Macintyre VG, Dickson JM, Fox JRE, Taylor PJ. Experiences of shame and guilt in anorexia and bulimia nervosa: a systematic review. Psychol Psychotherapy: Theory Res Pract. 2020;93:134–59. https://doi. org/10.1111/papt.12198.
- van Dyck Z, Lutz APC. On the relationship between body perception and eating disorders in adolescents and young adults. In: Heinen A, Samuel R, Vögele C, Willems H, editors. Wohlbefinden Und Gesundheit Im Jugendalter. Wiesbaden: Springer VS; 2022. https://doi.org/10.1007/978-3-658-35744-3_15.
- Loeb KL, Bernstein KS, Dimitropoulos G. What does weight have to do with atypical AN? A commentary on weight outcomes for adolescents with atypical anorexia nervosa in family-based treatment. PubMed. 2023;32(3):172–6.
- Harrop EN, Hutcheson R, Harner V, Mensinger JL, Lindhorst T. You don't look anorexic: atypical anorexia patient experiences of weight stigma in medical care. Body Image. 2023;46:48–61. https://doi.org/10.1016/j. bodyim.2023.04.008.
- MacNeil BA, Nadkarni P, Leduc S, Leung PC. Hurry up and Weight: innovative inter-professional Outpatient Care for a Service user with severe Anorexia Nervosa awaiting admission to (Specialized Elective) Inpatient Care in Canada. Can J Community Mental Health. 2016;35(3):113–7. https://doi. org/10.7870/cjcmh-2016-040.
- Froreich FV, Vartanian LR, Grisham JR, Touyz SW. Dimensions of control and their relation to disordered eating behaviours and obsessive-compulsive symptoms. J Eat Disord. 2016;3(4):14. https://doi.org/10.1186/s40337-016-0104-4. PMID: 27144009; PMCID: PMC4853853.
- Gremillion H. Chapter Three: Hierarchy, Power, and gender in the Therapeutic Family. Feeding Anorexia: gender and power at a treatment Center. Duke University Press; 2003. pp. 119–56.
- LaMarre A, Levine MP, Holmes S, Malson H. An open invitation to productive conversations about feminism and the spectrum of eating disorders (part 1): basic principles of feminist approaches. J Eat Disorders. 2022;10(1). https://doi.org/10.1186/s40337-022-00532-x.
- LaMarre A, Levine MP, Holmes S, Malson H. An open invitation to productive conversations about feminism and the spectrum of eating disorders (part 2): potential contributions to the science of diagnosis, treatment, and prevention. J Eat Disorders. 2022;10(1). https://doi.org/10.1186/s40337-022-00572-3.
- Holmes S, Malson H, Semlyen J. Regulating untrustworthy patients: Constructions of trust and distrust in accounts of inpatient treatment for anorexia. Feminism Psychol. 2021;31(1):41–61. https://doi. org/10.1177/0959353520967516.
- Reay M, Holliday J, Stewart J, Adams J. Creating a care pathway for patients with long standings, complex eating disorders. J Eat Disorders. 2022;10(128). https://doi.org/10.1186/s40337-022-00648-0.
- Public Health Agency of Canada. Social Determinants of Health. Government of Canada, Public Health Agency of Canada, Canadian Best Practices Portal. 2016. https://cbpp-pcpe.phac-aspc.gc.ca/public-health-topics/social-determinants-of-health/# Accessed 21 Feb 2023.
- Richards PS, Caoili CL, Crowton SA, Berrett ME, Hardman RK, Jackson RN, Sanders PW. An exploration of the role of religion and spirituality in the treatment and recovery of patients with eating disorders. Spiritual Clin Pract. 2018;5(2):88–103. https://doi.org/10.1037/scp0000159.
- Heiden-Rootes K, Linsenmeyer W, Levine S, Oliveras M, Joseph M. A scoping review of research literature on eating and body image for transgender and nonbinary youth. J Eat Disorders. 2023;11(1):168. https://doi.org/10.1186/ s40337-023-00853-5.
- Weisz C, Quinn DM. Stigmatized identities, psychological distress, and physical health: intersections of homelessness and race. Stigma Health. 2018;3(3):229–40. https://doi.org/10.1037/sah0000093.
- Springmann ML, Svaldi J, Kiegelmann M. Theoretical and Methodological Considerations for Research on eating disorders and gender. Front Psychol. 2020;11. https://doi.org/10.3389/fpsyg.2020.586196.
- 63. Nasser M, Malson H. Beyond western dis/orders: thinness and self-starvation of other-ed women. In: Malson H, Burns M, editors. Critical feminist approaches to eating dis/orders. Routledge; 2009. pp. 74–83.
- Duong D, Vogel L. Overworked health workers are past the point of exhaustion. CMAJ. 2023;195(8):E309–10. https://doi.org/10.1503/cmaj.1096042.
- Favaro A. Dr. Judy Morris Canada's Health Workforce Crisis. [Audio podcast episode]. In Canadian Health Information Podcast; 2022. https://www.cihi.ca/

- $en/pod cast/dr-judy-morris-canadas-health-workforce-crisis \ Accessed\ 20\ Jan\ 2024.$
- Tse A, Xavier S, Trollope-Kumar K, Agarwal G, Lokker C. Challenges in eating disorder diagnosis and management among family physicians and trainees: a qualitative study. J Eat Disorders. 2022;10(45). https://doi.org/10.1186/ s40337-022-00570-5.
- Obeid N, Coelho JS, Booij L, Dimitropoulos G, Silva-Roy P, Bartram M, et al. Estimating additional health and social costs in eating disorder care for young people during the COVID-19 pandemic: implications for surveillance and system transformation. J Eat Disorders. 2024;12(1). https://doi. org/10.1186/s40337-024-01003-1.
- Statistics Canada. Mental health care needs, 2018 [Internet]. Statcan.gc.ca. Government of Canada, Statistics Canada. 2019. https://www150.statcan.gc.ca/n1/pub/82-625-x/2019001/article/00011-eng.htm
- 69. Obeid N, Silva-Roy P, Booij L, Coelho JS, Dimitropoulos G, Katzman DK. The financial and social impacts of the COVID-19 pandemic on youth with eating disorders, their families, clinicians and the mental health system: a mixed

- methods cost analysis. J Eat Disorders. 2024;12(1). https://doi.org/10.1186/s40337-024-00986-1.
- Stone KD, Dimitropoulos G, MacMaster FP. Food for Thought: A Dissonance Between Healthcare Utilization Costs and Research Funding for Eating Disorders in Canada. Journal of the Canadian Academy of Child and Adolescent Psychiatry [Internet]. 2021;30(3):197–203. https://www.ncbi.nlm.nih.gov/ pmc/articles/PMC8315219/
- Jennings Mathis K, Anaya C, Rambur B, Bodell LP, Graham AK, Forney KJ, Anam S, Wildes JE. Workforce diversity in the field of eating disorders: a Multi-methods Study. West J Nurs Res. 2020;42(12):1068–77. https://doi. org/10.1177/0192945920912396.

Publisher's note

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.